\documentclass[11pt,twoside]{article}

\usepackage{AR99}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\addtolength{\topmargin}{-0.5in}
\addtolength{\textheight}{0.5in}
\def\baselinestretch{0.93}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\authorfoot{E. Franconi, I. Horrocks}
\shorttitle{DLs with n-ary Relations}
\title{Description Logics with n-ary Relations}
\author{Enrico Franconi and Ian Horrocks\\
   Dept. of Computer Science, Univ. of Manchester \\
  Manchester M13 9PL, UK \\
  ${\tt \{{\tt franconi}\mid{\tt horrocks}\}@cs.man.ac.uk}$\\
  ${\tt http://www.cs.man.ac.uk/\{{\tt \sim\!franconi}\mid{\tt \sim\!horrocks}\}}$
}
\date{}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\usepackage{dl}

%% Style file for creating Grail/description logic documents

% Necessary Packages

\usepackage{xspace}
\usepackage{amssymb}

% Hyphenation

\hyphenation{data-base tab-leaux tab-leau}

% Abreviations

\newcommand{\KB}{KB\xspace}
\newcommand{\KBfull}{\textrm{knowledge base}\xspace}
\newcommand{\TKB}{TKB\xspace}
\newcommand{\TKBfull}{\textrm{terminological knowledge base}\xspace}
\newcommand{\DL}{\textrm{DL}\xspace}
\newcommand{\DLs}{\textrm{DLs}\xspace}
\newcommand{\DLfull}{\textrm{Description Logic}\xspace}
\newcommand{\DLsfull}{\textrm{Description Logics}\xspace}

% Acronyms etc.

\newfont{\mymathtt}{cmtt10 scaled 1095}
\newcommand{\ttt}[1]{\mbox{\mymathtt #1}}

\newcommand{\scname}[1]{\textsc{#1}}%   a (small) capitalised name
\newcommand{\calig}[1]{%                caligraphic text
        \ensuremath{\mathcal{#1}}}
\newcommand{\grail}{\scname{Grail}\xspace}%     GRAIL
\newcommand{\Grail}{\grail}%                    GRAIL
\newcommand{\galen}{\scname{Galen}\xspace}%     GALEN
\newcommand{\Galen}{\galen}%                    GALEN
\newcommand{\alc}{%                             ALC
  \ensuremath{\mathcal{ALC}}\xspace}
\newcommand{\alcx}[1]{\calig{ALC#1}}%           ALCx
\newcommand{\alctrans}{%                        ALC_+
        \ensuremath{\alc_{+}}\xspace}
\newcommand{\alcfi}{\ensuremath{\mathcal{ALCFI}_{R^+}}\xspace}
\newcommand{\alcRtrans}{%                       ALC_R+
        \ensuremath{\alc_{R^{+}}}\xspace}
\newcommand{\alcOtrans}{%                       ALC_o+
        \ensuremath{\alc_{\oplus}}\xspace}
\newcommand{\alcXtrans}[1]{%                    ALCx_+
        \ensuremath{\alcx{#1}_{+}}}
\newcommand{\alchrp}{%                          ALCH_R+
  \ensuremath{\mathcal{ALCH}_{R^{+}}}\xspace}
\newcommand{\alchfrp}{%                         ALCHf_R+
  \ensuremath{\mathcal{ALCH}\textbf{\textit{f}}_{R^{+}}}\xspace}
\newcommand{\Fact}{FaCT\xspace}%                FaCT
\newcommand{\Classifier}{\Fact}%
\newcommand{\Dlp}{DLP\xspace}%                  DLP
\newcommand{\Tsl}{\calig{TSL}\xspace}%          TSL
\newcommand{\Ciq}{\calig{CIQ}\xspace}%          QIC
\newcommand{\Back}{\scname{Back}\xspace}%       BACK
\newcommand{\Classic}{\scname{Classic}\xspace}% CLASSIC
\newcommand{\Loom}{\scname{Loom}\xspace}%       LOOM
\newcommand{\Kris}{\scname{Kris}\xspace}%       KRIS
\newcommand{\Ksat}{\scname{Ksat}\xspace}%       KSAT
\newcommand{\Crack}{\scname{Crack}\xspace}%     CRACK
\newcommand{\PDL}%                              PDL
        {PDL\xspace}
\newcommand{\CPDL}%                             converse-PDL
        {\emph{converse}-PDL\xspace}
\newcommand{\Lisp}{\textrm{Lisp}\xspace}
\newcommand{\Prolog}{\textrm{Prolog}\xspace}
\newcommand{\Smalltalk}{\textrm{SmallTalk}\xspace}

% Semantics etc.

\newfont{\bigmathxx}{cmsy10 scaled 1440}
\newfont{\smallmathxx}{cmsy10 scaled 720}

\newcommand{\bigsqcap}{\mathop{\mathop{\mbox{\bigmathxx\symbol{117}}}}\limits}
\renewcommand{\bigsqcup}{\mathop{\mathop{\mbox{\bigmathxx\symbol{116}}}}\limits}

\newcommand{\I}{%                               I (caligraphic)
        \ensuremath{\mathcal{I}}\xspace}
\newcommand{\Tau}{%                             Tau
        \calig{T}\xspace}
\newcommand{\Tree}{%                            Tree
        \ensuremath{\mathbf{T}}\xspace}
\newcommand{\ifunc}{%                           ^I (caligraphic)
        \ensuremath{^\mathcal{I}}\xspace}
\newcommand{\deltai}{%                          Delta^I
        \ensuremath{\Delta\ifunc}\xspace}
\newcommand{\Exptime}{\textsc{Exptime}-complete\xspace}
\newcommand{\Pspace}{\textsc{Pspace}-complete\xspace}

\newcommand{\Term}{%                            Terminology
        \ensuremath{\mathcal{T}}\xspace}

% Algorithms

\newcommand{\Concepts}{\ensuremath{\mathbf{C}}\xspace}
\newcommand{\Roles}{\ensuremath{\mathbf{R}}\xspace}
\newcommand{\Rplus}{\ensuremath{\mathbf{R}_+}\xspace}
\newcommand{\Rp}{\ensuremath{\mathbf{R}_\mathsf{P}}\xspace}
\newcommand{\Rfunc}{\ensuremath{\mathbf{F}}\xspace}
\newcommand{\Lab}{\ensuremath{\EuScript{L}}\xspace}
\newcommand{\Edges}{\ensuremath{\EuScript{E}}\xspace}
\newcommand{\Inv}{\mathop{\mathsf{Inv}}}
\newcommand{\Tr}{\mathop{\mathsf{Trans}}}

% Modal Logics

\newcommand{\K}%                                Modal K
        {\ensuremath{\mathbf{K}}\xspace}
\newcommand{\KT}%                               Modal KT
        {\ensuremath{\mathbf{KT}}\xspace}
\newcommand{\Kfour}%                            Modal K4
        {\ensuremath{\mathbf{K4}}\xspace}
\newcommand{\Sfour}%                            Modal S4
        {\ensuremath{\mathbf{S4}}\xspace}
\newcommand{\Km}%                               Modal Km
        {\ensuremath{\K_{(\mathbf{m})}}\xspace}
\newcommand{\KTm}%                              Modal KTm
        {\ensuremath{\KT_{(\mathbf{m})}}\xspace}
\newcommand{\Kfourm}%                           Modal K4m
        {\ensuremath{\mathbf{K4}_{(\mathbf{m})}}\xspace}
\newcommand{\Sfourm}%                           Modal S4m
        {\ensuremath{\mathbf{S4}_{(\mathbf{m})}}\xspace}


% Concept names etc.

\newcommand{\key}[1]{%                  a key word
        \ensuremath{\texttt{#1}}}
\newcommand{\con}[1]{%                  a concept
        \ensuremath{\textsf{#1}}}
\newcommand{\role}[1]{%                 a role
        \ensuremath{\textsf{\textsl{#1}}}}
\newcommand{\attr}[1]{%                 an attribute/feature/functional role
        \role{#1}}
\newcommand{\ind}[1]{%                  an individual
        \con{#1}}
\newcommand{\coneqn}[1]{%               a concept equation
        \con{#1}}
\newcommand{\cname}[1]{\con{#1}}%       a concept name e.g. human
\newcommand{\rname}[1]{\role{#1}}%      a role name
\newcommand{\aname}[1]{%                an attribute name
        \cname{#1}}


% DL syntax & notation

\newcommand{\Subs}{\ensuremath{\sqsupseteq}}
\newcommand{\Issub}{\ensuremath{\sqsubseteq}}
\newcommand{\nSubs}{\ensuremath{\not\sqsupseteq}}
\newcommand{\nIssub}{\ensuremath{\not\sqsubseteq}}
\newcommand{\Tail}{\mathop{\mathsf{Tail}}}

\newcommand{\some}[2]{%
        \ensuremath{\exists #1 . #2}}
\newcommand{\all}[2]{%
        \ensuremath{\forall #1 . #2}}
\newcommand{\atleast}[2]{%
        \ensuremath{\mbox{$\geqslant$}#1 #2}}
\newcommand{\atmost}[2]{%
        \ensuremath{\mbox{$\leqslant$}#1 #2}}
\newcommand{\atleastq}[3]{%
        \ensuremath{\mbox{$\geqslant$}#1 #2 . #3}}
\newcommand{\atmostq}[3]{%
        \ensuremath{\mbox{$\leqslant$}#1 #2 . #3}}
\newcommand{\feat}[1]{\ensuremath{(\leqslant 1 \; #1)}}
\newcommand{\Negfeat}[1]{\ensuremath{(\geqslant 2 \; #1)}}
\newcommand{\ain}[2]{%
        \ensuremath{#1 . #2}}
\newcommand{\aeq}[2]{%
        \ensuremath{#1 = #2}}
\newcommand{\aneq}[2]{%
        \ensuremath{#1 \neq #2}}
\newcommand{\aeqd}[2]{%                                 attributes eq and defined
        \ensuremath{#1 \stackrel{\downarrow}{=} #2}}
\newcommand{\aneqd}[2]{%                                attributes neq and defined
        \ensuremath{#1 \stackrel{\downarrow}{\neq} #2}}
\newcommand{\domain}[1]{%                               domain of a function
        \ensuremath{\mathit{dom}~#1}}
\newcommand{\id}[1]{%                                   identity relation
        \ensuremath{\mathit{id}(#1)}}
\newcommand{\trans}[1]{%                                transitive role operator
        \ensuremath{#1^{+}}}
\newcommand{\transref}[1]{%                             transitive-reflexive role operator
        \ensuremath{#1^{*}}}
\newcommand{\restr}[2]{%                                role restriction
        \ensuremath{#1\!\mid_{#2}}}
\newcommand{\tuple}[2]{%                                binary tuple
        \ensuremath{\langle #1 , #2 \rangle}}
\newcommand{\constraint}[2]{%                           constraint x:C
        \ensuremath{#1\textrm{:}#2}}
\newcommand{\Card}[1]{%                                 set cardinality
        \ensuremath{\vert#1\vert}}
\newcommand{\Trole}{%                                   transitive role
        \ensuremath{^{(+)}}\xspace}


% non-LaTex composed symbols

\newcommand{\dotin}{%
        \ensuremath{\dot{\in}}}
\newcommand{\dotneq}{%
        \ensuremath{\dot{\neq}}}


% GRAIL syntax & notation

\newcommand{\gnewsub}[2]{%
        \ensuremath{#1\  \key{newSub}\  #2}}
\newcommand{\gcrit}[2]{%
        \ensuremath{#1\ #2}}
\newcommand{\gwhich}[2]{%
        \ensuremath{#1~\key{which}~#2}}
\newcommand{\gname}[2]{%
        \ensuremath{#1\  \key{name}\  #2}}
\newcommand{\gtnecc}[2]{%
        \ensuremath{#1\  \key{topicNecessarilly}\  #2}}
\newcommand{\gnewind}[2]{%
        \ensuremath{#1\  \key{newIndividual}\  #2}}
\newcommand{\greally}[2]{%
        \ensuremath{#1\  \key{really}\  #2}}


% Other stuff

\newcommand{\Eg}{\textrm{e.g.}\@\xspace}
\newcommand{\Ie}{\textrm{i.e.}\@\xspace}

\newcommand{\sss}{{\mathrel{\kern.25em{\sqsubseteq}\kern-.5em \mbox{{\scriptsize *}}\kern.25em}}}
\newcommand{\R}{\ensuremath{\mathcal{R}}\xspace}
\newcommand{\alci}{\ensuremath{\mathcal{ALCI}_{R^+}}\xspace}
\newcommand{\alchi}{\ensuremath{\mathcal{ALCHI}_{R^+}}\xspace}
\newcommand{\alch}{\ensuremath{\mathcal{ALCH}_{R^+}}\xspace}
\newcommand{\Sat}{\emph{satisfiable}\xspace}
\newcommand{\Unsat}{\emph{unsatisfiable}\xspace}

% Note - the following is provided by the llncs class

%\newcommand{\qed}{{\unskip\nobreak\hfil\penalty50
%   \hskip2em\hbox{}\nobreak\hfil
%   \qedd
%   \parfillskip=0pt \finalhyphendemerits=0
%    \medskip\goodbreak\noindent}}
\newcommand{\qedd}{\vrule height4pt width 4pt depth0pt}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\DLR}   {\mbox{${\cal DLR}$}}
 
\newcommand{\Ifact}{i\Fact}%                I-FaCT
\newcommand{\TopC}{\ensuremath{\top_\mathcal{C}}}
\newcommand{\TopT}{\ensuremath{\top_\tau}}
\newcommand{\Topn}{\ensuremath{\top_n}}

\renewcommand{\rmdefault}{ptm} 
\renewcommand{\sfdefault}{phv}
\renewcommand{\ttdefault}{pcr}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{CDR98}

\bibitem[CDR98]{Calvanese98a}
D.~Calvanese, G.~{De~Giacomo}, and R.~Rosati.
\newblock A note on encoding inverse roles and functional restrictions in
  $\mathcal{ALC}$ knowledge bases.
\newblock In E.~Franconi, G.~{De~Giacomo}, R.~M. MacGregor, W.~Nutt, C.~A.
  Welty, and F.~Sebastiani, editors, {\em Collected Papers from the
  International Description Logics Workshop (DL'98)}, pages 69--71. CEUR, May
  1998.

\bibitem[DL99]{lenzerini:kr-98}
G.~{De Giacomo} and M.~Lenzerini.
\newblock Description logic framework for information integration.
\newblock In {\em Proceedings of the 6th International Conference on the
  Principles of Knowledge Representation and Reasoning (KR-98)}, pages 2--13.
  Morgan Kaufmann, 1999.

\bibitem[FS99]{franconi:sattler:aiia-99}
Enrico Franconi and Ulrike Sattler.
\newblock A data warehouse conceptual data model for multidimensional
  aggregation: a preliminary report.
\newblock {\em {AI*IA} Italian Association for Artificial Intelligence
  journal}, 1999.
\newblock To appear.

\bibitem[Hor98]{Horrocks98c}
I.~Horrocks.
\newblock Using an expressive description logic: {FaCT} or fiction?
\newblock In A.~G. Cohn, L.~Schubert, and S.~C. Shapiro, editors, {\em
  Principles of Knowledge Representation and Reasoning: Proceedings of the
  Sixth International Conference (KR'98)}, pages 636--647. Morgan Kaufmann
  Publishers, San Francisco, CA, June 1998.

\bibitem[HPS99]{Horrocks98i}
I.~Horrocks and P.~F. Patel-Schneider.
\newblock Optimising description logic subsumption.
\newblock {\em Journal of Logic and Computation}, 1999.
\newblock To appear.

\bibitem[HS99]{Horrocks98j}
I.~Horrocks and U.~Sattler.
\newblock A description logic with transitive and inverse roles and role
  hierarchies.
\newblock {\em Journal of Logic and Computation}, 1999.
\newblock To appear.

\bibitem[HST99]{Horrocks99b}
I.~Horrocks, U.~Sattler, and S.~Tobies.
\newblock Practical reasoning for expressive description logics.
\newblock Submitted, 1999.

\end{thebibliography}


\maketitle

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this abstract we will briefly mention the research work which has
been done at the University of Manchester on recent advances in
Description Logics (DLs), namely the \DLR\ generalisation of DLs
towards n-ary relations. \DLR\ was first introduced
by~\cite{lenzerini:kr-98} as a means for encoding expressive semantic
data models for information systems such as extended Entity
Relationship.  It turns out that \DLR\ elegantly generalises standard
DLs, by allowing formulae referring to n-ary relations, not just to
binary relations (i.e., roles). Still, the flavour of the obtained
language is a modal logic, in the sense that formulae state properties
holding in classes of worlds, determined by means of possibly complex
paths constructed from n-ary relations. As a special case of the
language, we again obtain a standard DL or a standard normal modal
logic.
%
We present here the first order fragment of the language introduced
in~\cite{lenzerini:kr-98}. We have chosen to limit the expressivity to
first order since we are looking for a language implementable with the
current technology.  In particular, we refer to the current academic
implementations of expressive DLs, namely the systems
\Fact~\cite{Horrocks98c} and \Ifact (\Fact extended with inverse
relations).  It has been recently demonstrated~\cite{Horrocks98i} that
the logic we are considering here allows for the implementation of
sound and complete reasoning algorithms that behave quite well both in
realistic applications and systematic tests.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The semantics of \DLR\ rely on a model theoretic {\em interpretation\/}
$\I = (\Delta^\I,\cdot^\I)$, consisting of a set $\Delta^\I$, called
the {\em domain\/}, and an interpretation function $\cdot^\I$ that
assigns to each concept $C$ a subset $C\ifunc$ of \deltai and to each
relationship \role{R} of arity $n$ a subset $\role{R}\ifunc$ of
$(\deltai)^n$, such that the conditions in column~2 of
Table~\ref{tab:concepts}{} are satisfied (where $A$ is an atomic
concept, \role{P} is an atomic relationship, $C$ and $D$ are concepts,
$R$ and $S$ are relationships, $\$i$ denotes the $i$th component of an
$n$-ary relationship, and $k$ is a non-negative
integer).


\begin{table}%[t!]
\begin{center}
{\setlength{\arraycolsep}{0em}
\begin{tabular}{|l|l|l|}
\hline
\DLR\ syntax & \DLR\ semantics & DL encoding\\
\hline
%
\TopC &
$\TopC\ifunc \subseteq \deltai$ &
$\TopC \quad (\TopC \sqsubseteq \top)$ \\
%
$A$ &
$A\ifunc \subseteq \TopC\ifunc$ &
$A \quad (A \sqsubseteq \TopC)$ \\
%
$\neg C$ &
$\TopC\ifunc \setminus C\ifunc$ &
$\neg C$ \\
%
$C \sqcap D$ &
$C\ifunc \cap D\ifunc$ &
$C \sqcap D$ \\
%
$\exists[\$i]\role{R}$ &
$\{d \in \TopC\ifunc \mid \exists\langle d_1,\ldots,d_n\rangle 
\in \role{R} . d_i = d\} \leqslant k\}$ &
\some{U_i^-}{C_R} \\
%
$\leqslant k [\$i]\role{R}$ &
$\{d \in \TopC\ifunc \mid \sharp\{\langle d_1,\ldots,d_n\rangle 
\in \role{R} \mid d_i = d\} \leqslant k\}$ &
\atmostq{k}{U_i^-}{C_R} \\
%
\hline
%
\Topn &
$\Topn\ifunc \subseteq (\deltai)^n$ &
$\Topn \quad (\Topn \sqsubseteq \top)$ \\
%
\role{P} &
$\role{P}\ifunc \subseteq \Topn\ifunc$ &
$C_P \quad (C_P \sqsubseteq \Topn)$ \\
%
$\neg\role{R}$ &
$\Topn\ifunc \setminus \role{R}\ifunc$ &
$\neg C_R$ \\
%
$\role{R} \sqcap \role{S}$ &
$\role{R}\ifunc \cap \role{S}\ifunc$ &
$C_R \sqcap C_S$ \\
%
$\$i_n:C$ &
$\{\langle d_1,\ldots,d_n\rangle \in \Topn\ifunc \mid d_i \in C\ifunc\}$ &
$\Topn \sqcap \all{U_i}{C}$ \\
%
\hline
\end{tabular}}
\end{center}
\caption{\DLR\ concepts and relationships, their semantics and 
  their DL encodings}\label{tab:concepts}
\end{table}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

It turns out that first order \DLR\ theories can be encoded into an
expressive DL theory, equipped with generalised axioms, inverse
relationships, and graded modalities (i.e., qualified number
restrictions)~\cite{lenzerini:kr-98}. A \DLR\ theory consists of a
finite set of axioms of the form $C \sqsubseteq D$ and $R \sqsubseteq
S$, where $C$ and $D$ are concepts, and $R$ and $S$ are relationships
of equal arity. Such a theory can be encoded in an equisatisfiable DL
theory using the encodings described in column~3 of
Figure~\ref{tab:concepts} (where $C_R$ is a concept corresponding to
the \DLR\ relationship $R$, and $U_i$ is a functional role
representing the $i$th component of a \DLR\ relationship).  Note that
\DLR\ relations are reified as DL concepts, and the components of the
relationship represented as functional roles.  While the encoding
obviously relies on the ability of the DL to reason with inverse
roles, another less obvious consequence is that this DL does not have
the finite model property. New techniques have been studied which
extend the algorithm of the \Fact system with blocking strategies that
allow complete reasoning with such an expressive
DL~\cite{Horrocks98j}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The \DLR\ Description Logic was designed such that it is able to
capture database schemata expressed in the most interesting Semantic
Data Models and Object-Oriented Data Models
\cite{lenzerini:kr-98,franconi:sattler:aiia-99}. For that purpose, the
first order fragment suffices.  The most common conceptual data model
for database and information system design is the Entity-Relationship
(ER) semantic data model.  \cite{lenzerini:kr-98} refers to an
extended version of the basic ER model, called EER, including
taxonomic relationships, arbitrary boolean constructs, and entity
definitions by means of both necessary and sufficient conditions and,
more generally, by means of generalised axioms. It is shown how a
schema expressed in an EER conceptual data model can be expressed in a
\DLR\ theory -- whose models correspond with legal database states of
the EER schemata -- allowing for reasoning services such as
satisfiability of a schema or the computation of a logically implied
formula.

Various experiments have been done in the context of the European
ESPRIT-4 Long Term Research project ``Foundations of Data Warehouse
Quality'' (DWQ), using EER schemata as a source for testing the
efficiency of reasoning with \DLR\ theories in realistic domains.
Previous approaches~\cite{Calvanese98a} based on the translation of
first order \DLR\ theories into the basic \alc DL (in correspondence
with multi-modal \mbox{$\cal K$}) introduce very large numbers of
axioms, and have proved impractical.  Using \Ifact and a slightly
simplified version of the above encoding, promising initial results
have been obtained with a \DLR\ theory representing (fragments of)
extended real world database schemata.  Work is in progress to extend
the algorithm implemented in \Ifact in order to allow a complete
encoding of \DLR~\cite{Horrocks99b}).

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\bibliography{aisb99-bib}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\end{document}
